Author's response to reviews

Title: Dietary and other lifestyle correlates of serum folate concentrations in a healthy adult population in Crete, Greece: a cross-sectional study

Authors:

Christos M Hatzis (kafatos@med.uoc.gr)
George K Bertsias (gbert@med.uoc.gr)
Manolis Linardakis (linman@med.uoc.gr)
John M Scott (jscott@tcd.ie)
Anthony G Kafatos (kafatos@med.uoc.gr)

Version: 3 Date: 29 December 2005

Author's response to reviews: see over
Dear Dr. Kumagai,

Attached please find the revised version of the manuscript **MS: 1853228168825706** (“Dietary and other lifestyle correlates of serum folate concentrations in a healthy adult population in Crete, Greece: a cross-sectional study”), to be considered for publication in the Nutrition Journal. The abstract has been reviewed by a native English speaker and has been edited so that “Methods” and “Results” are included in the same page.

We look forward to hearing from you.

The Authors.